Author's response to reviews

Title: Individual and family environmental correlates of television and computer time in 10- to 12-year-old European children: the ENERGY-project.

Authors:

Maïté Verloigne (maite.verloigne@ugent.be)
Wendy Van Lippevelde (wendy.vanlippevelde@ugent.be)
Elling Bere (elling.bere@uia.no)
Yannis Manios (manios@hua.gr)
Eva Kovacs (kovacs.eva.katalin@pte.hu)
Monika Grillenberger (mgrillenberger@web.de)
Lea Maes (lea.maes@ugent.be)
Johannes Brug (j.brug@vumc.nl)
Ilse De Bourdeaudhuij (ilse.debourdeaudhuij@ugent.be)

Version: 2
Date: 28 May 2015

Author's response to reviews: see over
Dear Editor,

We have made the changes in the manuscript entitled “Individual and family environmental correlates of television and computer time in 10- to 12-year-old European children: the ENERGY-project” (Maïté Verloigne, Wendy Van Lippevelde, Elling Bere, Yannis Manios, Kovács Éva, Monika Grillenberger, Lea Maes, Johannes Brug, Ilse De Bourdeaudhuij), a research paper.

Following request was made:

(1) Please provide the specific names of all ethics committees which granted approval in the manuscript.

Answer:

We have added all ethics committees in the manuscript:

*The study protocol was approved by the ethics committee of each country. The study was approved by the Medical Ethics Committee of the University Hospital Ghent in Belgium; by the State Medical Chamber of Baden-Württemberg in Germany; by the Bioethics Committee of Harokopio University in Greece; by the Scientific and Ethics Committee of Health Sciences Council in Hungary; and by the National Committees for Research Ethics in Norway.*
We apologize for this lack of information in the first version and we hope our manuscript is now in good order and will be considered as a valuable contribution to “BMC Public Health”.

Sincerely yours,

Maïté Verloigne
Ilse De Bourdeaudhuij

**Corresponding Author:**

Maïté Verloigne

Ghent University, Faculty of Medicine and Health Sciences, Department of Movement and Sports Sciences, Watersportlaan 2, 9000 Gent, Belgium

Tel: +32 9 264 63 23, Fax:+32 9 264 64 84

Maite.Verloigne@Ugent.be